Author’s response to reviews

Title: IL-6 signaling by STAT3 participates in the change from hyperplasia to neoplasia in NRP-152 and NRP-154 rat prostatic epithelial cells

Authors:

Beverly E. Barton (bartonbe@umdnj.edu)
Thomas F. Murphy (Murphytf@umdnj.edu)
Patricia Adem (Adempa@umedj.edu)
Richard A. Watson (Watsonra@umdnj.edu)
Robert J. Inwin (Irwinri@umdnj.edu)
Hosea F. Huang (huanghf@umdnj.edu)

Version: 6 Date: 30 Oct 2001

1) Figure 5 was changed to show the effects of dexamethasone on high and low IL-6-expressing clones of NRP-154 cells.
2) The AG490 data include the effects on high and low IL-6-expressing clones of NRP-154 cells.
3) The text reflects these changes, in the appropriate places.
4) The text was changed to conform with all the specific comments of the reviewer.